Author’s response to reviews

Title: The impact of study design and diagnostic approach in a large multi-centre ADHD study: Part 2: Dimensional measures of psychopathology and intelligence.

Authors:

Ueli C Müller (u.c.mueller@bluewin.ch)
Philip Asherson (Philip.Asherson@iop.kcl.ac.uk)
Tobias Banaschewski (tobias.banaschewski@zi-mannheim.de)
Jan K Buitelaar (J.Buitelaar@psy.umcn.nl)
Richard P Epstein (Epstein@netvision.net.il)
Jaques Eisenberg (Jaques@cc.huji.ac.il)
Michael Gill (mgill@tcd.ie)
Iris Manor (imanor@clalit.org.il)
Ana Miranda (Ana.Miranda@uv.es)
Robert D Oades (robert.oades@uni-due.de)
Herbert Roeyers (Herbert.Roeyers@UGent.be)
Aribert Rothenberger (arothen@gwdg.de)
Joseph A Sergeant (JA.Sergeant@psy.vu.nl)
Edmund JS Sonuga-Barke (eib3@soton.ac.uk)
Margaret Thompson (M.Thompson@soton.ac.uk)
Stephen V Faraone (sfaraone@childpsychresearch.org)
Hans-Christoph Steinhausen (steinh@kjpd.uzh.ch)

Version: 6 Date: 26 March 2011

Author’s response to reviews:

- The abstract of the manuscript is pasted to the submission system
- Figure titles are inserted in the resubmitted manuscript
- No changes are made to the manner of citing additional files, because - after already having changed citation style in the previous submission - no violation of the rules mentionned in the editor’s e-mail could be identified. Please change the ways of citing additional files according to your specific rules.
- Typography has not been changed since last submission because already comprehensively checked.